Author's response to reviews

Title: Utilization and Cost of a New Model of Care for Managing Acute Knee Injuries: the Calgary Acute Knee Injury Clinic

Authors:

Breda Lau (bredalau@gmail.com)
Mark Lafave (mlafave@mtroyal.ca)
Nicholas Mohtadi (mohtadi@ucalgary.ca)
Dale Butterwick (butterwii@ucalgary.ca)

Version: 2 Date: 1 August 2012

Author's response to reviews:

RE: MS 1606342081772632

Dear Editors/Reviewers,

I have made the following revisions to the previously submitted manuscript:
1. I have added an ethics statement in the Methods section.
2. A competing interests section was created and placed after the conclusion.
3. An Author’s contribution section was placed after Competing interests.
4. An acknowledgment section was added after the author’s contribution section.

My apologies for not following the Instructions to the Author. I look forward to hearing from you.

Respectfully Yours,
Breda Lau